Author's response to reviews

Title: PDE4-targeted treatments for autoimmune diseases

Authors:

Neal Kumar (neal.kumar@tufts.edu)
Ari M Goldminz (ari.goldminz@tufts.edu)
Noori Kim (nkim@tuftsmedicalcenter.org)
Alice B Gottlieb (agottlieb@tuftsmedicalcenter.org)

Version: 3 Date: 14 February 2013

Author's response to reviews:

Formatting changes applied as requested below:

Please remove all issue numbers from the reference list.

Please reformat the references so that the title and volume are given in bold and the journal name in italics.

Please note that each reference can contain up to 30 authors' names for each citation. The term "et al." should not be used if there are fewer than 30 authors.

Please provide a list of all abbreviations used in the manuscript, formatted in to a paragraph. Please make sure all abbreviations are spelt out and occur in alphabetical order.

References 78 and 79 do not appear in the reference list. Please check this and correct appropriately.